Weekly Status Week of October 30 – November 5, 2008

Day of Year 304 - 310
Mission Day 2908 - 2914
Earth Observing-One (EO-1) General

EO-1 has continued to perform a 2 week long experiment that started 27 October 2008 and will end 9 November 2008 to characterize the dark current drift on the ALI instrument for use by the Landsat Data Continuity Mission development team.  Consequently, there were only 6 Data Collection Events (DCEs) scheduled this week.  

NOTICE:

There is planned an 8th anniversary celebration commemorating the launch of EO-1 that occurred on November 21, 2000.  The celebration will take place on Tuesday, November 25 at GSFC in conference room C432 in Bldg 23 from 3:30 to 6:00.  Light refreshments will be served.  
INSTRUMENTS

All instruments operated nominally this week.

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders. Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

On Monday, November 3, a meeting was held with the Defense Intelligence Agency (DIA), National Reconnaissance Office (NRO), and Department of Homeland Security (DHS) to discuss the status and plans for the cross-domain Sensor Web activity.  The following Action Items were generated at the meeting:

1.
Steve Schmitz to send ENVI Spear Tools to Nick Eftimiades to declassify and provide to Pat 



Capppelaere.
2.
Nick Eftimiades to identify universities/scientists working on hyperspectral products to help with 
classifiers and National Signatures Program (NSP) library usage.
3.
Nick Eftimiades to follow up with Todd Hawley & Naval Postgraduate School working on 
signatures.  Nick to look into potential funding for end user representatives such as they that can utilize cross-domain capabilities and provide feedback.
4.
DIA to get USAFRICOM stamp of approval of needs.
5.
GSFC (Mandl/Policelli) to provide Nick ground sensor requirements for forecasting models.
6.
Nick to look into the Shadow Harvest C-130 aircraft tasking for African flood scenario.
7.
GSFC & Al to give Estel feedback on white paper.
8.
Pat to give Nick flood products details for January Demo (Completed).
9.
DIA to give GSFC more information on Spectral Users' Group (SMUG).
10.
Kevin Fisher - Wrapping Automated Material Detection (AMD) as a Web Service: Open Issue - What's the plan?

11.
DIA to give GSFC classified access to signature's library to decide on what to use for flood demo 
(Ungar/Policelli/Kerekes/Hyde).  Location & security level TBD.
12.Fort Detrick National Center for Medical Intelligence is to be contacted for flood products 

(disease
vectors, historical data, etc.).
13.DHS/GSFC - Identify need/applicability for SensorWeb to participate in Interagency Remote 

Sensing Cell.
14.DHS to observe technology demonstration and applicability for NORTHCOM/FEMA domestic 

applications (our next demo in summer 09).
15.GSFC to provide DHS access to SensorWeb KML data.
16.NRO to provide Security Guard delivery schedule and work with DIA on operational Security 


Guard to use for January demonstration.
17.Dan Mandl - Deliver real products to real users (Red Cross/SERVIR) for a real upcoming flood 

disaster in Africa in January.
18.Steve Ungar to provide possible next hyperspectral technology demonstration Mission/


Instrument information and trade-offs to Pat.
19.Steve Schmitz to provide Laser Comm Downlink Bandwidth informationj to Pat.
20.DIA to identify intermediate milestones to accelerate SensorWeb effort.

21.DIA to meet with Raytheon regarding Hyperspectral capability (Techsat-3 ARTEMIS) and 

provide information to group.
22.Next meeting is Dec 1st at GSFC at 9:00AM.
A teleconference between the Advanced Information System Technology (AIST) Sensor Web collaborators was held on Thursday, October 30.  The following items of interest were discussed:

· Participants:
David Smithbauer, Steve Kolitz, Rob Sohlberg, Stu Frye, Dan Mandl, Pat Cappelaere, and Joe Young

· There was a discussion about various spectral band-based formulas used for creating burn products.  Dave Smithbauer has implemented a JPL formula and will send it to Rob Sohlberg.  Rob Sohlberg uses a Forest Service formula.  Use of multiple formulas may be valuable.
· Dan Mandl reported that at the recent HyspIRI meeting there was explained the need for additional on-board processing, to obtain information on 740 Mbps downlink capability, and investigate direct broadcast.  Rob Wright, Simon Hook, Dar Roberts, and Dan Mandl combined to push for these added capabilities so that HyspIRI can accommodate Sensor Web functions for fire and volcano rapid response disciplines.
· Dar Roberts has produced a fire potential index and map that shows a particular index for a particular region. 

· There will be another HyspIRI science team meeting by August 2009.

· Meeting with Rob Sohlberg and Mark Carroll to discuss their work on a global MODIS water mask is set for November 13.

· The upcoming ESTO AIST final review for the Supplemental Award was discussed and the schedule for the team’s input was set.

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week.  

EO-1 MISSION OPERATIONS CENTER

Real-Time

Spacecraft state of health is currently nominal.

EO-1 has been conducting calibration studies to measure the impact of imaging frequency on ALI sensor background noise by taking measurements of the ALI dark current drift.

· EO-1 performed a single Data Collection Event (DCE) per day between October 27 and November 2.
· EO-1 since has been performing a single DCE every other day.

· This calibration effort will be completed on November 9.

Primary T&C machine:
· Primary commanding and telemetry machine has been rebuilt.

· It is currently shadowing all contacts.

· FOT will complete installation of all newly developed lights-out automation software.

· FOT will conduct parallel operations between the primary and backup real-time machines before flagging the Primary T&C machine as operational.

· FOT is conducting real-time operations on the backup T&C.

· All network requirements for the backup T&C now complete.

· Planning to upgrade both machines to have the latest ASIST version and the operating system 
Development of the new Automated Notification System: “Event Monitor” is now complete.

New Goal Monitor 

· FOT has been successfully conducting lights-out automation with the new Goal Monitor.

· Software is still under development to perform additional tasks.

· UEO ground stations’ scheduling file deliveries.

· Prioritizing between requests based on the “latest uplink” times.

Mission Planning

ASPEN upgrade:

· Compiler machine placed on CNE network and successfully passed all IT security scans.

Flight Dynamics

PF2 tracking data problem resolved:
· Initial delivery of the geodetic data supplied by the station was incorrect.
· FOT requested updated geodetic data.
· Residuals from the orbit determination calculations disappeared immediately after implementing the new geodetics.

GROUND AND SPACE NETWORK

Station Downtimes

McMurdo Ground Station (MGS), McMurdo, Antarctica:
· Station status set to RED until further notice due to scheduled maintenance.

· FOT will be granted replacement polar contacts from other GN stations.
Operational Discrepancies

· No major problems to report.
Upcoming Events

Momentum management tests:

· FOT and JPL will conduct tests to better understand the behavior of the reaction wheels during DCE related slews.

· Wheel speeds and rates will be trended to generate a profile.

· Certain momentum management constraints might be relaxed based on these studies to allow for increased DCE(s) during any given orbit.

Lunar Calibration:

· An all instrument nominal lunar calibration will be performed on Thursday, Nov. 13th during the 16:28z umbra.

· A modified lunar calibration will be performed three orbits following the nominal calibration.

MLT control maneuvers:

· FOT will conduct inclination maneuvers to increase MLT to 10am.

· First maneuver will begin at 14:02:06z on Dec 2, 2008.

· Second maneuver will begin at 13:47:13z on Dec4, 2008.

· Both maneuvers will be 400 sec. in duration.

Imagery Status

Scenes and Engineering Cals planned for week of October 30 – November 5, 2008              6
Total scenes and engineering calibrations planned for entire mission – approximately 42,757
Total Scenes:  ALI scenes in the Level 0 archive            38,634 (as of November 5, 2008)

                        Hyperion scenes in the Level 0 archive    38,384
Publications and Presentations Status 

404 publications 

284 external presentations

53 articles and press releases
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